Author's response to reviews

Title: Care seeking behaviour and barriers to accessing services for sexually transmitted infections among female sex workers in Laos

Authors:

Ketkesone Phrasisombath (ketkesone@hotmail.com)
Sarah Thomsent (sarah.thomsen@ki.se)
Vanphanom Sychareun (vsychareun@gmail.com)
Elisabeth Faxelid (elisabeth.faxelid@ki.se)

Version: 2 Date: 2 June 2011

Author's response to reviews:

The corresponding author has made the suggested revision to the manuscript including:

1. The title page which contains the name, institutions, countries, email addresses and full postal address of all authors is added in the front page of the manuscript.
2. The abstract had changes according to the guidelines of the journal.